Author's response to reviews

Title: Study Protocol: Systematic Review and Meta-Analysis of Randomized Controlled Trials in First-Line Treatment of Squamous Non-Small Cell Lung Cancer

Authors:

Amy M DeLozier (delozier_amy@lilly.com)
Jacqueline Brown (jackie.brown@lilly.com)
Fanni Natenagara (natanegara_fanni@lilly.com)
Luping Zhao (zhao_luping@lilly.com)
Zhanglin Lin Cui (cui_zhanglin@lilly.com)
Stephen L Able (able_stephen_l@lilly.com)
Lee Bowman (bowman_lee_lb@lilly.com)
Joseph Treat (treat_joseph_anthony@lilly.com)
Lisa M Hess (hess_lisa_m@lilly.com)

Version: 4 Date: 3 September 2014

Author's response to reviews:

Response to comments
1) Please include a competing interests section at the end of the manuscript, before the reference list. If the authors have no competing interests, please state: "The authors declare that they have no competing interests."

We have included a notation that all authors are employees of Eli Lilly and company. This is disclosed on the cover page with our affiliations.

2) If applicable, please include an acknowledgement section at the end of the manuscript before the reference list. Please acknowledge anyone who contributed towards the study by making substantial contributions to conception, design, acquisition of data, or analysis and interpretation of data, or who was involved in drafting the manuscript or revising it critically for important intellectual content, but who does not meet the criteria for authorship. Please also include the source(s) of funding for all authors. Authors should obtain permission to acknowledge from all those mentioned in the Acknowledgements. Please state clearly whether or not you have funding in the acknowledgement section. If there is no funding, please state this.

We do not include any acknowledgements as all the work was done by those listed as authors. We have included a statement of funding.

3) Please include an Authors? Contributions section at the end of the manuscript, before the reference list. Each author should be listed individually. We suggest the following kind of format (please use initials to refer to each author's contribution): ?AB carried out the molecular genetic studies, participated in the
sequence alignment and drafted the manuscript. JY carried out the immunoassays. MT participated in the sequence alignment. ES participated in the design of the study and performed the statistical analysis. FG conceived of the study, and participated in its design and coordination and helped to draft the manuscript. All authors read and approved the final manuscript.

This has been added to the manuscript as requested.